Author's response to reviews

Title: Activation of transcription factor AP-1 in response to thermal injury in rat small intestine and IEC-6 cells

Authors:

Hong Zhao (zhaohong2556@126.com)
Yonghong Zhang (yunfengzhao@126.com)
Tao Liu (liutaoshuier168@163.com)
Changrong Wan (changrongwan@126.com)
Xiaoxi Liu (liuxiaoxi_06@163.com)
Zhimin Gao (gaozhimin123@126.com)
Xiaolin Hou (xlhsx@163.com)
Fenghua Liu (Liufenghua1209@126.com)

Version: 3 Date: 27 June 2015

Author's response to reviews:

1. We have revised the manuscript in light of referee 2's comment.
2. All section and subsection numbers were removed.
3. The Acknowledgement section was moved.
4. We have revised the manuscript in accordance to the journal style.